Notes from Pixel Mechanics Meeting

May 14, 2001

Present: Anderssen, McCormack, Taylor, Wirth, Gilchriese, Hartman, Johnson, Dardin, Weber

1. Mockup status.

The alignment of the tube mockup is continuing. Neal to look with Steve. Drawings for the frame mockup have been started but put at lower priority for now compared to SCT interfaces and other work on support tube design. 

2. Fitting status

Tom W. presented a spreadsheet that partially summarizes the testing done to date. He and John T. will update this and circulate to this mailing list.  The rough summary(from my perspective) is

Glued fittings work thru 8 bar test after irradiation in C3F8, but glue really looks bad(although seems to work fine). Pictures are on Fred's site, sufficiently ugly that one would have to understand source. 

Variseals irradiated in C3F8 look OK through 8 atm test if appropriate-sized fittings are used. 

Indium seal designed by Fred needs more compressive force on In(actually less In). Redesign.

Lure-lok type fittings are OK through 8 atm test and irradiation is not relevant since metal to metal.

Brazed samples were tested with variseals. Before irradiation they look OK, although one braze joint of 5 leaked. Seal surface is rough(apparently from flux) even after UHV clean, although seal is OK. Brazing on back burner unless laser welding flames out.

Revision to assembly to enable cold He test has been completed and reaches -20C. Good enough to get moving on these tests.

Contacts with laser welders have been on hold during Neal's absence. Neal will find out about delivery of parts from EB and contact Laser Fab directly. 

3. "Clean room"

Jon has to move out of former clean room in 2 weeks since it will be demolished. Will relocate sector assembly to "glass shop" near his office. 

4. Support tube design status

General status will be summarized in slides for meeting on Wednesday by Gil. Will send these when done.

5. Stave testing

Thermal cycling of irradiated stave has started in "by hand" fashion from room T to -35F and back. Should be ready for imaging next week.

Work continues on "vortex tube" for cooling, have reached about -20C. More work will continue.

In addition, will set up LN2 dewar near imager and try cooling old sector with cold vapor to simulate thermal shock.  If this works, then do stave.

Other stave testing outlined in note to Tom W, not repeated here.

